Author's response to reviews

Title: Effects of air pollution on neonatal prematurity in Guangzhou of China, a time-series study

Authors:

Qingguo Zhao (zqgfrost@126.com)
Zhijiang Liang (liangzhijiang07@163.com)
Shijuan Tao (victoriataojian@yahoo.cn)
Juan Zhu (litte-nurse@163.com)
Yukai Du (duyukai100@yahoo.com.cn)

Version: 5 Date: 16 December 2010

Author's response to reviews:

Dear editor

Thank you for your suggestions on our manuscript.

This time, we had a fluent English writer to edit the grammar, and adjusted the formatting according to instruction for author. All changes made were highlighted in red color.

Anyway, the figure1 has been revised according to request of photograph, and will be submitted separately. Please check them. Thank you again.

Best regards

Yours sincerely Zhao Qingguo